\documentclass[10pt]{article}
\usepackage[colorlinks=true, linkcolor=blue, urlcolor=blue]{hyperref}

\begin{document}

\title{Pagoda Bibliography Test File}
\author{Pagoda}
\date{Today}
\maketitle

\section{UPC++ citations}

\cite{upcxx-ipdps19}
\cite{upcxx-paw17}
\cite{upcxx-site}
\cite{upcxx-spec}
\cite{upcxx-guide}
\cite{upcxx-paw21-kokkos}

\section{GASNet citations}

\cite{gasnet-lcpc18}
\cite{gasnet-pawatm22}
\cite{gasnet-sc21-kinds}
\cite{gasnet-aries}
\cite{gasnet-site}
\cite{gasnet-spec-1.8.1}
\cite{gasnet-spec-1.1}

\newpage
\bibliography{pagoda}
\bibliographystyle{habbrv}
\end{document}

